Author's response to reviews

Title: Treatment of progressive supranuclear palsy with autologous adipose tissue-derived mesenchymal stem cells: a case report

Authors:

Soo Won Choi (love575dr@naver.com)
Kwon Byong Park (naskbpark@hanmail.net)
Sang Kyu Woo (paulskwoo@rnl.co.kr)
Sung Keun Kang (kangsk@rnl.co.kr)
Jeong Chan Ra (jcra@rnl.co.kr)

Version: 5
Date: 27 November 2013

Author's response to reviews: see over
Author’s response to reviews

Title: Treatment of progressive supranuclear palsy with autologous adipose tissue-derived mesenchymal stem cells: a case report

Authors:

Soo Won Choi (love575dr@naver.com)
Kwon Byong Park (nskbpark@hanmail.net)
Sang Kyu Woo (paulskwoo@k-stemcell.com)
Sung Keun Kang (kangsk@k-stemcell.com)
Jeong Chan Ra (jcra@k-stemcell.com)

Date: 27 November 2013

Author’s response to reviews: see over
Thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your reviewer’s comments.

1. Please change the description of the patient’s gender from man to male?.

   Done

2. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

   Page 2, Line 30, We have revised from “South Korean” to “Asian male from South Korea”.
   Page 5, Line 82: from “A man” to “An Asian male”.

3. Please remove the name of the hospital where the patient was confined, as this may jeopardize the anonymity of the patient.

   Done

4. Please remove the dates of confinement in the Case Presentation section.

   Done

5. Please include an acknowledgement section at the end of the manuscript before the reference list.

   We have included the ACKNOWLEDGEMENT section.
Comment:

We have revised the E-mail address for some authors (Sang Kyu Woo, Sung Keun Kang, Jeong Chan Ra) in title page because the company name changed from “RNL BIO” to “K-STEMCELL” at July 30, 2013. All changes were highlighted.

Thank you again for your time and consideration.

Your sincerely.